Author’s response to reviews

Title: Over-indebtedness as a marker of socioeconomic status and its association with obesity: a cross-sectional study

Authors:

Eva Muenster (eva.muenster@uni-mainz.de)
Heiko Rueger (heiko.rueger@uni-mainz.de)
Elke Ochsmann (eochsmann@ukaachen.de)
Stephan Letzel (letzel@uni-mainz.de)
Andre Michael Toschke (michael.toschke@kcl.ac.uk)

Version: 3 Date: 5 June 2009

Author’s response to reviews:

Dear editors

Thank you very much for your email from 1st June 2009 and your invitation to resubmit a further version of our paper. We changed the paper according to the reviewer’s suggestion:

Reviewer: Yannis Manios

Reviewer’s report:
The authors have responded to reviewers’ comments adequately. However, there is one minor essential revision that needs to be made.

Following both the comments by Dr Plachta-Danielzik and myself, I think that the references 5-8, should be placed in the following sentence “the link between socioeconomic status and health is well documented, particularly for overweight” (Page 4, Paragraph 3) and not where they are currently placed.

ANSWER:
Thank you for your thorough review that is highly appreciated. We completely agree with this comment and have to apologise for the bug in our reference programme and have corrected this in the amended version of the paper.